Author's response to reviews

Title: Opportunities for Human Resources for Health and Rehabilitation: A Response to Jesus et al.

Authors:

Jessica Power (odowdje@tcd.ie)
Joanne Mcveigh (jmcveigh@tcd.ie)
Brynne Gilmore (gilmorb@tcd.ie)
Malcolm MacLachlan (mlachlan@tcd.ie)

Version: 1 Date: 15 Aug 2017

Author’s response to reviews:

Dear Editor,

We welcome the reviewers’ positive comments and constructive feedback.

We agree that the original paper by Jesus et al. is an important article, which draws attention to an understudied area. This response was meant to complement this paper by adding further elements taking a global perspective. We have added a sentence in this regard in the opening paragraph and this is also present in the closing paragraph.

While there are numerous references for a commentary piece, we found that as we were discussing key topic areas, it was necessary to support this with the key references in each of these areas. These support the importance and effectiveness of these approaches (e.g. governance, task-shifting, CBR and assistive technology). The bibliography also can provide readers with a useful resource for key literature in these fields.

We look forward to your response.

Best wishes,

Jessica Power